Author's response to reviews

Title: Reversal of childhood idiopathic scoliosis in an adult, without surgery: a case report and literature review

Authors:

William J Brooks (wibdo@wjbrooksdo.com)
Elizabeth A Krupinski (krupinski@radiology.arizona.edu)
Martha C Hawes (mhawes@u.arizona.edu)

Version: 9 Date: 9 December 2009

Author's response to reviews:

AUTHOR RESPONSES IN BOLDFACE TYPE, BELOW.

Formatting changes to be made:

-----------------------------

Major revisions (we require the author to make these changes):

Consent: Please include a consent statement including the following information 'Written informed consent was obtained from the patient for publication of this case report and accompanying images. A copy of the written consent is available for review by the Editor-in-Chief of this journal.' Although the patient was the third author, consent should still be explicitly included within the manuscript.

DONE, AS REQUESTED.

Authors' contributions: Please use the following format as an example, including using initials to refer to each author's contribution: AB carried out the molecular genetic studies, participated in the sequence alignment and drafted the manuscript. JY carried out the immunoassays. MT participated in the sequence alignment. ES participated in the design of the study and performed the statistical analysis. FG conceived of the study, and participated in its design and coordination.

DONE, AS REQUESTED.

Authors' contributions: Please indicate that all authors read and approved the final manuscript.

DONE, AS REQUESTED.

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above):


Please include the country on the affiliation details.
DONE, AS REQUESTED.

Green lines in table 1 - please ensure all table lines are formatted in black.
DONE, AS REQUESTED.

Figures 1-3: It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image. Our online figure guide contains full details for preparing files for submission and can be viewed here: http://www.biomedcentral.com/info/ifora/figures
DONE, AS REQUESTED.

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.
DONE.

Thank you
THANK YOU!